Author's response to reviews

Title: Socio-economic differences in self-reported insomnia and stress in Finland from 1979 to 2002: a population-based repeated cross-sectional survey.

Authors:

Kirsi M Talala (kirsi.talala@thl.fi)
Tuija P Martelin (tuija.martelin@thl.fi)
Ari H Haukkala (ari.haukkala@helsinki.fi)
Tommi T Härkänen (tommi.harkanen@thl.fi)
Ritva S Prättälä (ritva.prattala@thl.fi)

Version: 4 Date: 7 March 2012

Author's response to reviews:

Dear Editorial Office,

Concerning my revised manuscript 5.3.2012, I just noticed that original tables were missing from the manuscript, and therefore I'm submitting a new correct version. I'm sorry for this confusion!